Author’s response to reviews

**Title:** An Immunocompetent Individual Presenting with Plasmablastic Lymphoma In the Ano-Rectal Junction: A Case Report

**Authors:**

- Mayur Brahmania (mab977@mail.usask.ca)
- Thomas Sylwesterowic (tsylwesterowic@shaw.ca)
- Heather Leitch (drhleitch@providencehematology.com)

**Version:** 2  **Date:** 18 May 2010

**Author’s response to reviews:**

The title was changed

The number of references was reduced to 28

Formatting changes, as outlined in the instruction for authors, were made

Square brackets were used for references

Figures and a table was added after the reference section